

JGR Earth Surface

RESEARCH ARTICLE

10.1029/2020JF005827

Key Points:

- A nested ice sheet model was applied to the last deglacial retreat in the Amundsen Sea sector, Antarctica. Model outputs were compared with ice thinning history
- Similar response to forcing across Amundsen Sea Embayment, possibly resulting from similar (ocean) forcing mechanisms in early Holocene
- Improved representation of regionally varying Earth-ice-sealevel interactions may resolve model-data mismatch in timing of ice thinning

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. S. Johnson, jsj@bas.ac.uk

Citation:

Johnson, J. S., Pollard, D., Whitehouse, P. L., Roberts, S. J., Rood, D. H., & Schaefer, J. M. (2021). Comparing glacial-geological evidence and model simulations of ice sheet change since the Last Glacial Period in the Amundsen Sea sector of Antarctica. *Journal of Geophysical Research: Earth Surface, 126*, e2020JF005827. https:// doi.org/10.1029/2020JF005827

Received 3 AUG 2020 Accepted 10 MAY 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Comparing Glacial-Geological Evidence and Model Simulations of Ice Sheet Change since the Last Glacial Period in the Amundsen Sea Sector of Antarctica

Joanne S. Johnson¹, David Pollard², Pippa L. Whitehouse³, Stephen J. Roberts¹, Dylan H. Rood⁴, and Joerg M. Schaefer⁵

¹British Antarctic Survey, Cambridge, UK, ²Earth and Environmental Systems Institute, Pennsylvania State University, University Park, PA, USA, ³Department of Geography, Durham University, Durham, UK, ⁴Department of Earth Science & Engineering, Imperial College London, London, UK, ⁵Lamont-Doherty Earth Observatory, Columbia University, Palisades, NY, USA

Abstract Since the Last Glacial Maximum ~20,000 years ago, the Antarctic Ice Sheet has undergone extensive changes, resulting in a much smaller present-day configuration. Improving our understanding of basic physical processes that played important roles during that retreat is critical to providing more robust model projections of future retreat and sea-level rise. Here, a limited-area nested ice sheet model was applied to the last deglacial retreat of the West Antarctic Ice Sheet in the Amundsen Sea Embayment (ASE), at 5 km resolution. The ice sheet response to climate and sea-level forcing was examined at two sites along the flowlines of Pine Island Glacier and Pope Glacier, close to the Hudson Mountains and Mount Murphy respectively, and the simulated responses compared with ice sheet thinning histories derived from glacial-geological data. The sensitivity of results to selected model parameters was also assessed. The model simulations predict a broadly similar response to ocean forcing in both the central and eastern ASE, with an initial rapid phase of thinning followed by a slower phase to the modern configuration. Although there is a mismatch of up to 5,000 years between the timing of simulated and observed thinning, the modeling suggests that the upstream geological records of ice surface elevation change reflect a response to retreat near the grounding line. The model-data mismatch could potentially be improved by accounting for regional variations in mantle viscosity, sea-surface heights and basal sliding properties across the continental shelf.

Plain Language Summary During the last glacial period approximately 20,000 years ago, the West Antarctic Ice Sheet was much larger than today. Since then, it has undergone extensive changes such as retreat and thinning of its glaciers, and currently dominates the contribution to global sea-level rise from Antarctica. Glacial-geological studies undertaken in two mountain ranges in the region (the Hudson Mountains and Mt Murphy, close to Pine Island Glacier and Pope Glacier) provided records of the magnitude and speed of those changes. Here we compare those records with model simulations that predict response of the ice sheet to different forcing mechanisms (climate and sea-level) over the past 20,000 years. We find that the model predicts a similar response to forcing at both study sites, with initially fast thinning followed by a slower phase as the ice sheet shrinks to its modern shape and size. The similarity in behavior at both sites may be due to similar ocean forcing mechanisms affecting both glaciers. When we compare the model simulations with geological records of ice sheet change, we find a mismatch of up to 5,000 years. Improving how ice-Earth interactions are represented in the model could potentially resolve this mismatch.

1. Introduction

The largest change in the configuration of the Antarctic Ice Sheet in the recent geologic past has been its retreat from Last Glacial Maximum (LGM) extent over the past ~20,000 years. During this time, grounding lines have retreated hundreds of kilometers across the continental shelf to their modern positions (RAISED consortium, 2014). The basic physical processes that played important roles during that retreat in Antarctica, such as increasing ice flux across deepening grounding lines, will continue to determine further retreat into deep marine basins driven by future climate warming (Allevropoulos-Borrill et al., 2020; Arthern &



Williams, 2017; DeConto & Pollard, 2016; Feldmann & Levermann, 2015; Golledge et al., 2019). Thus, improving our understanding of the timing and pace of the last deglacial retreat is critical to providing more robust model projections of future retreat and sea-level rise. Records of changes in surface elevation of the ice sheet through the last deglacial retreat are archived within geological sequences and can be extracted using chronological dating techniques. When obtained from several locations at the present ice sheet margins and inland, such records provide constraints on the varying size and shape of an ice sheet through time, and are thus invaluable for validating and tuning predictive ice sheet models (Albrecht et al., 2020a, 2020b; Briggs et al., 2014; Lowry et al., 2020; Pollard et al., 2016). Here, glacial-geological records of ice sheet surface lowering during the past 20,000 years across the Amundsen Sea sector of the West Antarctic Ice Sheet are compared with model simulations of the rate and magnitude of the last deglacial retreat in the same region.

2. Background

Glacial-geological records of Antarctic ice sheet retreat since the Last Glacial Maximum (LGM) were comprehensively reviewed and synthesized in 2014 (RAISED consortium, 2014). These were deduced using methods such as radiocarbon dating of marine sediments, bathymetric and seismic surveys of submarine glacial bedforms, and cosmogenic nuclide surface exposure dating of terrestrial rock outcrops near modern ice margins. A number of subsequent studies have further improved our knowledge of last deglacial ice sheet surface elevation changes across the continent (e.g., Antarctic Peninsula: Glasser et al., 2014; Jeong et al., 2018; Amundsen-Bellingshausen sector: Johnson et al., 2017; Johnson et al., 2020; Ross Sea sector: Balco et al., 2019; Goehring et al., 2019; Jones et al., 2015; Smellie et al., 2017; Spector et al., 2017, 2019; Yokoyama et al., 2016 and others; East Antarctica: Jones et al., 2017; Strub et al., 2015; Weddell Sea sector: Bentley et al., 2017; Balco et al., 2017; Nichols et al., 2019 and references therein). Marine geological data show that grounding line retreat from the outer continental shelf largely occurred prior to the Holocene, between approximately 20 and 10 ka, driven primarily by rising global sea-levels associated with Northern Hemispheric deglaciation. Continent-wide patterns of ice sheet thinning prior to the Holocene are not yet well-known because terrestrial geological data that provide reliable constraints on thinning from 20 to 10 ka are few (see Figure 11, Small et al., 2019). In contrast, several studies have shown that many hundred meters of ice sheet thinning occurred very rapidly at some locations during the early- to mid-Holocene (e.g., Bentley et al., 2011; Johnson, et al., 2019a, 2020, 2014; Jones et al., 2015; Spector et al., 2017). Grounding line retreat to near-modern positions during that time was probably driven by warming of sub-surface ocean water and an associated increase in sub-ice shelf melting (Golledge et al., 2013; Mackintosh et al., 2011; RAISED consortium, 2014).

The Amundsen Sea sector of the West Antarctic Ice Sheet, in particular the Pine Island-Thwaites Glacier catchment, is presently experiencing ongoing rapid thinning and grounding line retreat (Konrad et al., 2018; Milillo et al., 2019; Rignot et al., 2014), which has increased five-fold in the past two decades (Shepherd et al., 2019). Ice sheet models suggest that this retreat may continue and accelerate in coming centuries, driven by Marine Ice Sheet Instability with increasing ice flux across deepening grounding lines, and/or by Marine Ice Cliff Instability with failure of tall ice cliffs (e.g., Arthern & Williams, 2017; DeConto & Pollard, 2016; Feldmann & Levermann, 2015; Golledge et al., 2019, 2014). Detailed knowledge of how this sector of the ice sheet varied in both extent and thickness in the last deglacial period is critical for understanding how it can respond to future climate change. A summary of the current state of knowledge is as follows: In the eastern ASE, by the early Holocene (10.6 ka), the grounding line had retreated from a position near the continental shelf edge at the LGM to the inner shelf (Smith et al., 2014). Between 12 and 10.6 ka, an ice shelf spanning Pine Island Bay buttressed the grounding line of Pine Island Glacier (Kirshner et al., 2012). High rates of Circumpolar Deep Water (CDW) upwelling onto the continental shelf during the early Holocene may have provided a mechanism for collapse of this ice shelf by 7.5 ka (Hillenbrand et al., 2017), and may also have been connected with rapid thinning of ice in the Hudson Mountains at \sim 8 ka (Johnson et al., 2014). In the central ASE, grounding line retreat occurred with a similar style and timing as in the east. The grounding line had retreated from its maximum LGM extent at the continental shelf edge to near its present location on the inner shelf by 10.1 ka (Smith et al., 2011) and several hundred meters of ice surface lowering occurred shortly after, between 9 and 6 ka (Johnson et al., 2020).





Figure 1. Map showing location of Hudson Mountains and Mt Murphy within the Amundsen Sea Embayment (ASE) (Note: map is rotated relative to inset). Ice surface speeds (Mouginot et al., 2012; Rignot et al., 2011a, 2017) are overlain on the Landsat Image Mosaic of Antarctica (Bindschadler et al., 2008). Bathymetry is from the International Bathymetric Chart of the Southern Ocean (Arndt et al., 2013). The grounding line (black) is from Rignot et al. (2011b), and the coastline (blue) from the Antarctic Digital Database version 7.1. Dashed yellow lines indicate approximate boundaries between outer (OUT), middle (MID) and inner (IN) shelf regions.

3. Rationale and Objectives

Although some modeling studies of the last deglacial retreat of the Antarctic ice sheet have examined specific regions, most have been concerned with large-scale behavior (Albrecht et al., 2020a, 2020b; Briggs et al., 2013, 2014; Golledge et al., 2014; Gomez et al., 2013, 2018; Lowry et al., 2020; Mackintosh et al., 2011; Maris et al., 2015; Pollard et al., 2016, 2017; Whitehouse et al., 2012). These studies have found broad-scale agreement with the observational pictures described above, but to our knowledge, no last-deglacial studies have focused exclusively on the ASE.

In this study, a nested ice sheet model spanning the ASE is applied to the last deglacial retreat of the Antarctic ice sheet to assess sensitivity to different environmental forcings, to investigate links between grounding line retreat and inland thinning, and to compare the predicted timing and magnitude of retreat in the eastern versus central ASE. The existing body of data described in Section 2 is used as context, but with a focus on comparisons with surface exposure age data from rock outcrops at two sites: the Hudson Mountains adjacent to Pine Island Glacier in the eastern ASE, and Mt Murphy adjacent to Pope Glacier in the central ASE (Figure 1). Constraints on past ice thicknesses at these sites (Johnson et al., 2014, 2020) provide useful limits on the magnitude of inland ice variations.

4. Methods

4.1. Compilation of Surface Exposure Age Data

In order to make a comparison between ice sheet thinning histories derived from terrestrial glacial-geological data and model simulations of thinning at the same sites in the ASE over the same last deglacial timescale (20,000 years to present), existing exposure age data were compiled from two key sites in the ASE (Figure 2). Exposure ages are calculated by measuring the abundance of cosmogenic nuclides that are produced





Figure 2. Compilation of exposure age data used for comparison with model simulations. Data are from Mt Murphy (Johnson et al., 2020) and the Hudson Mountains (Johnson et al., 2014). Error bars represent external errors, at 1 sigma. Dashed lines are modern ice surface elevations ('MM' for Mt Murphy, 'HM' for Hudson Mts). Best-fit thinning trajectories, based on Monte-Carlo linear regression analyses undertaken for the original studies, are shown as solid lines (black for Mt Murphy, magenta/green for Hudson Mts); see Supporting Information for method. The shaded bars represent the duration of the Holocene epoch and LGM.

by cosmic ray bombardment of an ice-free rock surface (or surface free of other glacial debris or sediments). In simple terms, the higher the measured abundance of the nuclide, the longer the surface has been ice-free. Thus, measurements in rock surfaces along an altitudinal transect act as a "dipstick" for the former height of an ice sheet surface as it thins through time (e.g., Stone et al., 2003). The Hudson Mountains and Mt Murphy exposure age datasets (Johnson et al., 2014, 2020) are detailed enough to provide evidence of several hundred meters of ice sheet surface lowering in the ASE during the last deglacial period (Figure 2). A few isolated exposure ages exist from elsewhere in the ASE (Figure 8, Johnson et al., 2020), but are not included in the compilation because they only provide a single deglaciation age at a particular site; several exposure ages would be needed to identify outliers. All exposure ages discussed in the present study were calculated using the online cosmogenic nuclide exposure age calculator, version 3 (Balco et al., 2008; hosted at https:// hess.ess.washington.edu). For ¹⁰Be, the calculations use the default global calibration data set, reference sea-level-high-latitude production rate $(3.92 \pm 0.31 \text{ at.g}^{-1}.yr^{-1})$, and the "LSDn" scaling scheme (Lifton et al., 2014), which yields smaller systematic biases for Antarctic sites than other commonly used scaling models. For in-situ ¹⁴C age calculations, the production rate is estimated from measurements of the CRO-NUS-A quartz inter-comparison standard (see Johnson et al., 2020), while all other variables are the same as for ¹⁰Be. For details of how the best-fit thinning trajectories were determined, see Supporting Information.



4.2. Ice Sheet Modelling

An ice sheet model (PSUICE3D) was applied to the last deglacial retreat of the Antarctic ice sheet. Model simulations are compared with trajectories for ice sheet surface lowering derived from the exposure age dating. The model is described further in the Supporting Information, and detailed formulations are given in Pollard & DeConto (2012a) and Pollard et al. (2015). This section describes aspects specific to this paper.

The model uses a polar stereographic grid centred on the South Pole, over a limited-area nested domain spanning the ASE to allow relatively fine horizontal resolution of 5 km. Model maps below show the entire nested domain. It is run from 30 ka to the present, with lateral boundary conditions obtained from a previous long-term continental-scale simulation at coarser resolution. The model is initialized at 30 ka from the continental-scale run to allow for ~10,000 years of spin up; results are shown here just for the last 20,000 years of the simulations. The continental-scale run (unpublished) is similar to that in Pollard & De-Conto (2009), except run from 125 to 0 ka at 20 km resolution with the current model version; it simulates LGM expansion and deglaciation to modern reasonably well, as in that paper. The long spin-up time, and the large buffer zones between the nested domain edges and the central region of interest, mean that the details of the continental run have very little impact on the results here.

Past atmospheric forcing is prescribed simply as uniform perturbations to modern climatology, as in previous applications (Pollard & DeConto, 2012a; Pollard et al., 2016, 2017). Air temperatures are obtained from a modern climatological Antarctic data set (Le Brocq et al., 2010), plus an imposed sinusoidal seasonal cycle, and spatially uniform past perturbations proportional to a deep-sea core δ^{18} O record (Lisiecki & Raymo, 2005) and orbital insolation variations. Lapse-rate corrections are applied to account for the difference between model and climate data set surface elevations. The observed annual mean precipitation (LeBrocq et al., 2010) is similarly corrected following temperature changes using a Clausius Clapeyron-like relation. The resulting temperature and precipitation variations over the last deglacial period agree well with those at the WAIS Divide core site (Cuffey et al., 2016; WAIS Divide Project Members, 2013; shown in Johnson et al., 2020, Figure 9d), both in timing and amplitude, and also with climate model output at that location (Lowry et al., 2019a, Figures 2 and 3).

Past oceanic forcing is prescribed from an archived atmospheric-oceanic global climate model (A/OGCM) simulation of the last 22 kyr with oceanic resolution of 3.6° longitude and $\sim 1.5^{\circ}$ latitude in high latitudes (Liu et al., 2009), using their water temperatures at 400 m depth corresponding roughly with CDW, spatially interpolated from their coarse global grid and extrapolated to the central ASE and under floating ice shelves (see Johnson et al., 2020, Figure 9b). A warming pulse occurs in the data set at 14 ka, with $\sim 0.2^{\circ}$ C running-mean amplitude and several hundred years duration. It corresponds in time with an imposed event in the Liu et al. (2009) experiment related to Meltwater Pulse 1A, and may be model dependent; it is removed in our forcing, but has negligible effect on our results. Global mean sea-level variations, driven mainly by Northern Hemispheric ice-sheet melting, are prescribed from the eustatic sea-level component of the ICE-5G data set (Peltier, 2004).

For the present study, a set of four simulations was used to show the basic range of model behavior over wide but reasonable variations of two uncertain model parameters: sub-ice oceanic melt coefficient (O; an overall factor multiplying the oceanic melt rate at the base of floating ice with nominal value 1 corresponding to K in Pollard & DeConto, et al., 2012a; Pollard et al., 2015) and e-folding time (τ ; yr) of as-thenospheric relaxation in the Elastic Lithosphere Relaxing Asthenosphere (ELRA) bed model (Pollard & DeConto, 2012a), as follows:

Simulation A: O = 1, τ = 3,000 Simulation B: O = 10, τ = 3,000 Simulation C: O = 1, τ = 100 Simulation D: O = 10, τ = 100

These two parameters and their ranges were chosen on the basis of previous continental-scale studies with much larger ensembles (Pollard et al., 2016, 2017); running such large ensembles for the ASE would be beyond the scope of this paper. The two parameters provide a preliminary exploration of the envelope of mod-



el behavior. Five additional simulations extending this exploration to other model parameters are described primarily in Supporting Information, but are also included in the discussion below:

Simulation E: O = 0, $\tau = 100$

Simulation F: O = 10, $\tau = 10$

Simulation G: O = 10, $\tau = 100$, alternate sea-level forcing (see Supporting Information)

Simulation H: O = 10, τ = 100, basal sliding coefficient on continental shelf $C_{\text{shelf}} = 10^{-7}$ (changed from 10^{-5}) m yr⁻¹ Pa⁻²

Simulation H': As H, except $C_{\text{shelf}} = 10^{-6} \text{ m yr}^{-1} \text{ Pa}^{-2}$

5. Results and Discussion

5.1. Variations in Ice Distribution and Velocity Between Simulations

Figure 3 shows maps of ice distribution for the two model simulations with the most and least ice retreat overall (B and C). Figure 3a shows maps of ice distribution at selected times for model simulation B, which yields the most realistic modern ice configuration of the four simulations tested, with D a close second (judged mainly by the differences from modern shown in Figure 4). In this simulation, there is rapid and widespread retreat of grounded ice over the outer and middle continental shelf in the central and eastern ASE between 14 and 13 ka. After 13 ka, grounding lines recede more gradually to their modern positions, with large ice shelves persisting over the inner- and mid-shelf regions until after 10 ka. This behavior and associated feedbacks are discussed in Section 5.2. Simulation C (Figure 3b) incorporates less ocean melt and faster bed rebound. Grounding line retreat is slower and later than in simulation B, with grounding lines still located in the mid-shelf region of the ASE at ~12 to 10 ka. This is in better general agreement with marine geological and geophysical records (Larter et al., 2014). However, at the end of simulation C, modern grounded and floating ice extents are considerably greater than observed, especially in the eastern ASE.

Figure 4 shows the modern state at the end of all four simulations. The most influential parameter for the modern state is the ocean melt coefficient (O); the simulations with larger values (B and D, with O = 10) produce relatively realistic modern ice distributions, whereas those with smaller values (A and C, with O = 1) produce modern ice extents that are unrealistically large. This is because, in all simulations, ocean melting becomes important later in the runs, during the Holocene (Section 5.2). Bed rebound is most important in the earlier pre-Holocene stages, so the influence of the asthenospheric relaxation time scale (τ) on the modern states in Figure 4 is small.

5.2. Glacier Response to Ocean Melt Coefficient and Isostatic Rebound

Figure 5 shows modeled ice thickness variations at the Hudson Mountains and Mt Murphy for all four simulations. Changes relative to present-day model thickness are shown, since these correspond directly to the changes recorded by the cosmogenic exposure age data. The model predicts a similar response to forcing of both Pine Island Glacier and Pope Glacier in all simulations, with an initially fast rate of ice surface lowering prior to the Holocene (i.e., before ~12 kyr BP), followed by a slower phase with only minor thinning through the Holocene to the present-day. In all simulations, most of the thinning occurs between 15 and 10 ka. This is much earlier than indicated by the glacial-geological data (Section 5.4).

5.2.1. Rapid Abrupt Episodes of Thinning Prior to the Holocene

The two simulations with large asthenospheric relaxation times (A and B) show rapid and abrupt thinning episodes at both sites simultaneously (~13–12 ka for A, ~14.5–13.5 ka for B; Figure 5). These episodes are also evident in other figures, including the map views for simulation B in Figure 3a. The prescribed sea-level rise, predominantly due to Northern Hemispheric ice melt occurring between ~15 and 10 ka (Peltier, 2004), may trigger these episodes. When sea-level rises sufficiently, it causes grounding lines to retreat rapidly across the relatively flat seafloor topography of the outer ASE continental shelf to submarine bedrock ridges of the mid shelf, which pin and stabilize the grounding line (Figure 7). This behavior is consistent with the widely recognized repeated pinning and unpinning of grounding lines between bedrock ridges during





Figure 3. Model maps for selected times (ka, thousands of years before present) during the last deglacial period. (a) For simulation B (O = 10, $\tau = 3,000$ years), (b) For simulation C (O = 1, $\tau = 100$ years). The grounding line is shown as a thick black line. Blue-to-white colors show grounded ice elevation landward of the grounding line (thin black lines between color intervals), and pink colors show floating ice thickness seaward of the grounding line (with no contours). Red stars denote the two field sites (Hudson Mountains and Mt Murphy).



Journal of Geophysical Research: Earth Surface



Figure 4. Maps of modern ice distribution, ice surface speeds, differences between modern modeled and observed ice elevations (model minus observed where ice is present in both, or \pm ice thickness where ice is present in only one), and areas where modeled grounded versus floating ice differs from observed (model grounded ice in brown, floating in cyan). The upper four rows show results at the end of model simulations A to D. The bottom row shows observed ice extents (Bedmap2, Fretwell et al., 2013) and ice speeds (Rignot et al., 2011a), with both datasets aggregated to the model's 5 km grid. The grounding line is shown by a thick black line, and the two field study sites are marked by red stars. For modern ice distributions, the contour scales for grounded (blue) versus floating (pink) ice can be distinguished as in Figure 3.

retreat, found in other modeling studies (e.g., Parizek et al., 2013; Seroussi et al., 2017), and is also suggested by the sudden jumps in grounding line positions in Figure 6, below.

The major thinning episodes are more sudden in simulations A and B than in C and D (Figure 5). This is because the slower bedrock rebound ($\tau = 3,000$ years in simulations A and B, compared with $\tau = 100$ years in C and D) under the retreating ice allows bedrock elevations at grounding lines to remain deeper for longer. This produces larger ice fluxes across the grounding lines (which depend strongly on local bedrock depth; Schoof, 2007; see Supporting Information), and consequently faster drawdown of interior ice. In simulations C and D with faster bedrock rebound, there is still widespread grounding line retreat and thinning, but it occurs more gradually, slowed by shallower grounding line depths and less ice flux from upstream (cf. Gomez et al., 2013, 2015). Nevertheless, most of the inland thinning at the study sites still occurs before ~10 ka.





Figure 5. Model variations of ice thickness (m) relative to modern during the past 20,000 years, at the grid points closest to two data sites: Hudson Mountains (eastern ASE) and Mt Murphy (central ASE). Panels show results for simulations with varying model parameters (see Section 4.2): O is a factor multiplying the sub-ice oceanic melt rate, and τ is the asthenospheric e-folding relaxation time in the ELRA bed model. (a) Simulation A (O = 1, τ = 3,000 years). (b) Simulation B (O = 10, τ = 3,000 years). (c) Simulation C (O = 1, τ = 100 years). (d) Simulation D (O = 10, τ = 100 years).



Figure 6. Thick lines: Model variations of ice thickness (m, left-hand axis) during the past 20,000 years, at the grid points closest to two data sites: (a) Hudson Mountains (eastern ASE), and (b) Mt Murphy (central ASE). Thin lines: grounding line position, downstream from its modern observed location (km, right-hand axis), along a prescribed transect roughly following a glacial flowline passing near each site (shown in Figure 7c). Blue: Simulation A (O = 1, τ = 3,000 years). Cyan: Simulation B (O = 10, τ = 3,000 years). Purple: Simulation C (O = 1, τ = 100 years). Orange: Simulation D (O = 10, τ = 100 years).



Figure 7. Model cross sections along transects corresponding roughly to glacial flowlines, for simulation D (O = 10, $\tau = 100$ years). (a) Passing close to the Hudson Mountains study site (eastern ASE, along the Pine Island basin and extending offshore along Pine Island Trough; cf. Figure 3, Larter et al., 2014). (b) Passing close to the Mt Murphy study site (central ASE, along Pope Glacier and straight across the continental shelf). Profiles of ice surface, ice base and bedrock elevations are shown for selected times during the past 15,000 years. For both (a) and (b), the *x*-axis shows distance downstream (km) from the start of the flowline. The *y*-axis shows elevation (m) relative to modern sea-level. The ocean surface is not shown, but would be horizontal lines extending to the right from the ice edge, at elevations given in the figure legend (m, relative to modern). The approximate positions of closest approach to the exposure age data sites are shown by vertical gray bars. Vertical dashed lines between labels IS, MS, and OS (inner, mid, and outer shelves, respectively) indicate the locations where the transects cross latitudes corresponding to the inner- to mid-shelf transition, mid- to outer-shelf transition, and outer-shelf edge. (c) Transect paths (brown lines, labeled A-B and A'-B'), superimposed on a map of modern observed ice speeds (Rignot et al., 2011a) aggregated to the model's 5 km grid, with the two field sites marked by red stars.

5.2.2. Holocene Thinning

After these major retreat episodes, the much slower retreat and ice thinning in the Holocene (post 10 ka) is likely forced by a combination of ocean warming and increased sub-ice-shelf melting, and the remaining minor amount of sea-level rise. Ocean temperatures in the model are prescribed using the archived A/OGCM (400 m) results of Liu et al. (2009), in which nearly all of the warming in the proximal ASE ocean occurs between 10 and 5 ka, with very little before 11 ka (see Johnson et al., 2020, Figure 9b). After 10 ka, global mean sea-level rises by only ~20 m (compared to a ~95 m rise between 20 and 10 ka; ICE-5G, Peltier, 2004), but this may play a role in helping to unpin grounding lines from the closely spaced ridges in the inner ASE (see Figure 7, below).

5.3. Ice Sheet Response Along Flowlines

The model results in Figure 6 show that the timing of inland thinning of grounded ice at both study sites (Hudson Mts and Mt Murphy) coincides with that of grounding line retreat, for both the large and relatively





Figure 8. Model variations of ice thickness (m) relative to modern model thickness during the past 20,000 years, at the grid points closest to two data sites: (a) Hudson Mountains and (b) Mt Murphy. Panel (c) shows an expanded portion of (a) for the Hudson Mts site. Blue: Simulation A (O = 1, $\tau = 3,000$). Cyan: Simulation B (O = 10, $\tau = 3,000$). Purple: Simulation C (O = 1, $\tau = 100$). Orange: Simulation D (O = 10, $\tau = 100$). Exposure age data for each site (Johnson et al., 2014, 2020) are shown as black dots for comparison, and, in contrast to the data in Figure 2, are plotted relative to the present-day ice surface for comparison with the model outputs (i.e., elevations are height above the modern ice surface; ice surface heights used for Mt Murphy sample elevations are those in Figure 5, Johnson et al., 2020). Dashed lines show the best-fit thinning trajectories to the data, as in Figure 2 but for ice thickness changes, instead of altitudes. The labeled horizontal dashed lines show the amount of ice thickening, relative to present, required to submerge the highest currently exposed peak or ice-free location visited.

abrupt retreats before 10 ka, and for the more gradual later Holocene retreats. Figure 7 provides another view of the retreat along flowlines, showing model transects for selected timeslices through the deglacial period along Pine Island Glacier and Pope Glacier, and extended to the outer continental shelf. Results for simulation D are shown, which along with B produces quite realistic modern configurations (Figure 4), but the later retreat is in less conflict with data (Figures 5,6 and 8). Each transect passes close to the respective field sites. For both transects, the greatest extent of grounding line retreat, from the outer- to mid-continental shelf, occurs before 10 ka. This is consistent with the retreat histories in Figures 5–6; much of this retreat occurs in a single jump, especially for the Pope Glacier transect. As discusse for Figure 5, we consider this to be forced by sea-level rise causing retreat back to pinning points on the mid-shelf. Subsequent grounding line retreat from 10 ka to present is less extensive and more gradual overall, caused by ocean warming increasing sub-ice melting and thinning of floating ice shelves, and augmented slightly by the remaining sea-level rise (as discussed for Figure 5). This later retreat also tends to occur in steps, reflecting pinning on the succession of bathymetric ridges on the inner shelf of the ASE (Figure 7).

The concurrence in the model simulations of inland thinning of grounded ice and grounding line retreat suggests that there is a correlation between retreat and thinning at the data sites themselves. Similarly, within a given time interval, the amount of upstream thinning at the data sites closely correlates with the



distance over which the grounding line has retreated. This correlation has a theoretical basis in studies of waves of adjustment propagating inland from marginal perturbations, with propagation times of less than ~500 years over the first ~500 km for an East Antarctic flowline (Figure 2, Alley & Whillans, 1984). This result is important because exposure age data from upstream/inland sites are frequently utilized to infer changes hundreds of kilometers downstream without necessarily having evidence that upstream ice surface changes reflect changes in grounding line position. However, whilst the relationship between timing and distance of grounding line retreat and ice sheet thinning appears to hold along most transects shown in Figure 7, where topographic highs occur (such as in Figure 7b), it may be interrupted upstream of those peaks, resulting in the ice sheet response to downstream changes not being transmitted as far inland. Thus, while exposure age data from the upstream nunataks provide reliable information on the downstream timing of grounding line retreat in this study, the relationship between retreat and thinning modeled here may not be the same under different bathymetric or other boundary conditions elsewhere.

5.4. Model-Data Comparison

When compared with the ice sheet thinning history derived from the surface exposure age data, the model outputs imply a potential mismatch with some of the geological data, especially at Mt Murphy. The most obvious differences between the model simulations and the ASE cosmogenic exposure age data involves the timing of the retreat (see Figure 8, where the exposure age data are plotted together with the modeled thinning at the two data sites).

In the model, ice thinning at both sites mainly occurs prior to the Holocene, between ~ 15 and 10 ka. This is in reasonable agreement with the exposure age data from lower elevations in the Hudson Mountains (i.e., below \sim 90 m, Figure 8c); at higher elevations at that site, thinning appears to have been faster than in the model, but model-data comparison is limited both by the lack of exposure ages above 370 m asl and any geological constraints on the timing of thinning prior to the early Holocene (Figure 2). In contrast with the Hudson Mountains, exposure ages from Mt Murphy do not form a well-defined trend, with samples at the same elevation often yielding ages that differ by several thousand years. This scatter is most likely due to exposure of upstream bedrock surfaces (the source of erratic cobbles) for long periods prior to glacial erosion, resulting in the nuclide inventory accumulated during previous ice-free periods remaining at the time of erratic emplacement; such "nuclide inheritance" is common in Antarctica (e.g., Balco, 2011). Thus, the choice of ages to include in the regression analysis is left to expert judgment. Here, the published regression line for a subset of samples from Mt Murphy (Johnson et al., 2020; Supporting Information) is used as the best estimate for the timing and pace of thinning. Using different samples could lower the gradient of the thinning trend shown in Figures 2 and 8b, permitting a better fit to the model output, but this would not be geologically reasonable because the potential for nuclide inheritance at this site means that older ages are unlikely to represent the timing of deglaciation. Model-data comparison using the published regression line implies that most thinning at Mt Murphy occurred up to 5,000 years later than in the model, between ~ 10 and 5 ka (Figure 8b).

A number of additional factors might contribute to model-data mismatch, as follows (some are explored further in Supporting Information by additional simulations):

(1) The exposure ages may not be correct.

Exposure age calculations rely on knowing how the production rate of cosmogenic nuclides in rock surfaces varies with time, location and elevation. To apply any production rate to a particular sample site, it is necessary to scale it for the altitude and latitude of the sample site. Several production rate scaling-schemes exist for ¹⁰Be and in-situ ¹⁴C (the nuclides from which the exposure age data in this study were determined) (Balco et al., 2008). To mitigate the possibility of model-data mismatch resulting from incorrect exposure age calculations, we used the scaling-scheme that is generally considered the most appropriate for Antarctic sites (LSDn; Lifton et al., 2014). When other published scaling-schemes and production rates are used, the resulting exposure ages differ by no more than ~1.4 ka, which is within the external errors of the ages. Thus, we do not consider the choice of production rate/scaling scheme to be a plausible explanation for model-data mismatch.

Other factors, such as snow cover or cover by glacial till, may cause apparent exposure ages to be too young. For Mt Murphy and the Hudson Mountains, the impact of snow cover on exposure age calculations was assessed in the original publications (Johnson et al., 2020 and 2014, respectively). For samples with uncorrected exposure ages in the 0–16 ka range, cover by 100 cm of snow with density of 0.35 g cm⁻³ for 8 months of every year (a best estimate based on the authors' experience of snow accumulation at these sites during the austral summer and assuming greater, more persistent, accumulation during the winter) makes the ages no more than 1,900 years older. Alternatively, if a covering of till of 10 cm thickness with density of 1.8 g cm⁻³ is assumed, the difference is less than 2,000 years. Neither of these alone is sufficient to explain a model-data mismatch of several thousand years. We consider a situation where both occurred simultaneously—which could bring the ages close to the model thinning trajectory—extremely unlikely because no continuous accumulations of till have been observed in the area (the glacial deposits are scattered cobbles perched on scoured bedrock surfaces).

Glacial isostatic adjustment (GIA) can also have an impact on exposure age calculations because it causes the elevation of a sample site, relative to sea-level, to change through time. This in turn affects cosmogenic nuclide production. Using the online tool "iceTEA" (http://ice-tea.org/en/tools/correct-elevation-change/; Jones et al., 2019a), a GIA correction was applied to the exposure age data to assess the potential impact of post-exposure elevation change on the age calculations (see Supporting Information). When corrected for the effect of elevation change predicted by three GIA models (ICE-5G, Peltier, 2004; ICE-6G, Peltier et al., 2015; W12, Whitehouse et al., 2012), the exposure ages in the range 0–10 ka (i.e., when the last deglaciation occurred; Figure 2) were no more than 600 years older, and for 5–16 ka (the range of the majority of exposure ages from this region; Figure 2) they were less than 2,000 years older. This difference is not sufficient to explain a >2,000-year model-data mismatch. The combined effect of snow cover and GIA would make the exposure ages up to 3,900 years older (although without data on depth and persistence of winter snow accumulation at the sample sites, it is difficult to assess the likelihood of such a shift). This could explain a model-data mismatch of more than a couple of thousand years, but it would not be sufficient to explain a mismatch of as much as 5,000 years (e.g., Figure 8b).

(2) Ice thicknesses at the data sites could be influenced by local small-scale topography or other micro-climatic factors.

An example is described in Johnson et al. (2020), where the presence of a prominent windscoop near one of the study sites at Mt Murphy (Turtle Rock) appears to have resulted in deglaciation of that site approximately 4,000 years earlier than the timing of regional deglaciation. However, comparable features are absent from other Mt Murphy sites, and also from the Hudson Mountains. Conversely, the model results at the grid cell closest to each site could be anomalous and not representative of the other nearby cells; however, no such pronounced cell-to-cell variations are seen in the model. Thus small-scale local topographic and micro-climatic variations are not likely explanations for model-data mismatch.

(3) The prescribed climatological forcing and its timing may be fundamentally unrealistic.

Since eustatic sea-level over the last 20,000 years is relatively well known, with most sea-level rise occurring before 10 ka (Peltier, 2004), this component of the model forcing is probably realistic, assuming regional sea-level changes in the ASE are close to the global mean. Small changes to the sea-level forcing do not significantly change the overall results (see simulation G in Figure S2, Supporting Information). The atmospheric forcing (Section 4.2) is also sufficiently well-known that its uncertainties would only have small effects on the ice-sheet changes addressed here (cf. Lowry et al., 2019b). However, the prescribed oceanic temperatures from Liu et al. (2009) are more questionable. To our knowledge, Liu et al. (2009) and Lowry et al. (2019a) are the only applicable atmosphere-ocean model integrations through the last deglaciation to date, and these coarse-grid global ocean models may not capture important small-scale variations of CDW upwelling onto the ASE continental shelf. CDW influence is thought to have increased significantly after 10.4 ka until 7.5 ka (Hillenbrand et al., 2017); this increase is in rough agreement with the proximal oceanic temperature trends in Liu et al. (2009) (see Figure 9b, Johnson et al., 2020) but not with those further offshore in Lowry et al. (2019a) (see their Figures 7 and 8). In order to improve how small-scale changes in CDW upwelling in the forcing are captured in the present study, a record of Holocene ocean temperature changes in the ASE is needed. This is the subject of ongoing research, but results are not yet available. However, simulation E-with no oceanic melting at all-produces little overall change in the model ice thinning



and retreat, which is still dominated by sea-level rise at \sim 15 to 10 ka (Figure S2, Supporting Information). In summary, the data used for ocean temperatures may not adequately capture small-scale variations in CDW influence in the ASE region, but this is probably not the main cause of the model-data mismatch.

(4) Sensitivity of grounding line dynamics.

In the model, grounding line retreat and upstream thinning between ~ 15 and 10 ka are forced mainly by sea-level rise, and there is little subsequent response to the later ocean warming between ~ 10 to 5 ka. It is reasonable to question the model sensitivity to these forcings, since the response to sea-level in the model is strongly controlled by an uncertain parameterization of ice flux across the grounding line (Schoof, 2007), and its treatment of sub-ice-shelf oceanic melting is relatively simple. In larger-scale Antarctic simulations, fair agreement with data has been found with this and other ice sheet models (Golledge et al., 2014; Pollard & DeConto, 2009; Pollard et al., 2016), but scale-dependent shortcomings for smaller regions are possible. For instance, the horizontal grid size in the model, 5 km, may not adequately resolve the bathymetry of the Amundsen Sea shelf. Previous sensitivity modeling studies in this region have found that grounding line retreat may be influenced significantly by fine-scale ridges in basal topography, with intervals of pinning on each ridge alternating with rapid retreat between ridges. These require horizontal resolutions of several hundreds of meters to ~1 km for accurate simulations (Durand et al., 2011; Parizek et al., 2013; Seroussi et al., 2017), which are not feasible here. However, these studies focused on modern grounding line locations and retreat upstream from them, and their results may not necessarily apply to early grounding line retreat over the mid- to outer-continental shelf of the ASE where the bedrock topography is less rugged, with smooth sedimentary strata and fewer potential pinning points than the predominantly crystalline inner shelf region (Gohl et al., 2013; Graham et al., 2009).

(5) The model does not account for spatial variations in sea-surface height.

In the model used here, sea-level is simply prescribed as a spatially uniform elevation, following the eustatic sea-level component of the ICE-5G data set (Peltier, 2004). This does not account for sea-surface geoid distortions due to ice-ocean-solid Earth gravitational interactions or Earth rotational effects. Coupling an ice sheet model with a global Earth–sea-level model is needed to capture these effects; for instance, as an ice mass retreats, the reduced gravitational attraction causes shoaling of the nearby ocean, shallower grounding line depths, reduced ice fluxes, and slower ice retreat in a negative feedback (Gomez et al., 2013, 2015, 2018). In ensembles of deglacial Antarctic simulations coupled with a global Earth–sea-level model, Pollard et al. (2017) found that results agreed reasonably closely with those from the uncoupled model, suggesting use of the latter may be adequate for continent-scale experiments. On the other hand, it may be inadequate at the finer-scales of the present study: sea-surface heights in the ASE will have departed from the global mean during deglaciation, and may have differed spatially between the eastern and western sides of the basin, with potentially strong effects on local ice retreat that can only be captured by a high-resolution coupled model. Therefore, it is possible that improvements to representation of sea-surface height in the model may help to resolve the model-data mismatch.

(6) The bed model does not include the effects of spatial variations in Earth rheology.

The locally relaxing asthenosphere and elastic lithospheric plate of the ELRA model provides a rough approximation to the more realistic Earth deformation captured by global Earth models, and as mentioned above, yields results in other studies that agree reasonably well with those using global Earth models. However, recent geophysical evidence identifies very weak low-viscosity zones in the mantle under parts of West Antarctica, including the ASE (An et al., 2015; Barletta et al., 2018; Heeszel et al., 2016, O'Donnell et al., 2017). The fast bedrock rebound associated with very weak mantle viscosity can have a strong negative feedback on deglacial ice retreat: as the grounding line retreats, rapid bedrock rebound shallows local ocean depths, causing thinner ice at the grounding line, reduced ice flux from upstream, and slower drawdown of interior ice (Gomez et al., 2013, 2015). The short isostatic relaxation timescale in simulations C and D ($\tau = 100$ years; Figure 8), or even the extreme value of $\tau = 10$ years used in simulation F (Figure S2, Supporting Information), may be representative of the weakest mantle viscosities suggested by the recent data, but this parameter is applied uniformly over the model domain, and does not capture the observed spatial variations. It is possible that regional variations in Earth properties could have larger effects on the timing of local ice retreat within the ASE, especially when combined with regional variations in sea-surface

heights (discussed above), which together control ocean depths at grounding lines. Again, capturing these effects will require coupling with regional Earth–sea-level models that include lateral variations in Earth properties (Gomez et al., 2018).

(7) The modeled LGM ice sheet configuration influences subsequent retreat.

A recent study by Albrecht et al. (2020a, 2020b) compared an ensemble of continental Antarctic simulations over the last two glacial cycles, generated using the PISM ice sheet model, with various geological data in order to better constrain model parameter ranges. In contrast to the results of the present study, Albrecht et al. predict that most grounding line retreat occurred after 10 ka, including in the ASE sector, which concurs with the Holocene timing of thinning implied by the ASE exposure age data described above. This later timing of modeled grounding line retreat may be related to their greater LGM ice thicknesses predicted around coastal areas (Albrecht et al., 2020b). In contrast, large-ensemble studies with the present model (Pollard et al., 2016, 2017) found that relatively slippery bed properties are appropriate on continental shelves. This produces relatively thin ice in coastal areas during the LGM, making those regions sensitive to sea-level forcing and susceptible to early retreat. Better constraints on LGM ice thicknesses would help to resolve model discrepancies and refine model parameters (see Sections 5.5.3 and 6, below).

To examine the potential for thicker LGM ice to resolve the model-data mismatch in timing of ice thinning, two additional simulations H and H' were performed with reduced sliding coefficients on the continental shelf (see Supporting Information). In these, the main model thinning is significantly delayed at both data sites, and simulation H improves the fit to the exposure age data at Mt Murphy (Figure S2). However, modeled modern ice thicknesses are unrealistic in these simulations, with far too thick ice at the data sites; also, initial grounding line retreat across the outer ASE occurs much later than indicated by geological data (see Text S2, Figures S3 and S4, Supporting Information). Nevertheless, the effects of the two simulations are different at the two data sites (Figure S2), which suggests that simulations with spatially varying sliding coefficients could could help to resolve the model-data mismatch.

In summary, we conclude that the most likely contributing factors to the model-data mismatch are that (a) the model does not capture regional variations in sea-surface heights, or strong variations in mantle viscosity within the ASE, and (b) a uniform basal sliding coefficient is prescribed for the whole continental shelf, missing potentially important spatial variability.

5.5. Wider Implications

5.5.1. Mechanism for Ice Sheet Thinning in the Central ASE

Upwelling of warm CDW onto the inner shelf of Pine Island Bay increased during the early Holocene (10.4– 7.5 ka; Hillenbrand et al., 2017) and may have triggered grounding line retreat and collapse of an ice shelf in the eastern ASE, resulting in rapid ice surface lowering in the Hudson Mountains (Johnson et al., 2014). As yet, however, there are no marine geological data available from the inner shelf in front of Pope Glacier (i.e., immediately north of the Crosson Ice Shelf; Figure 1) that could show whether a similar effect was experienced there. Nevertheless, the existing exposure age datasets from Mt Murphy and the Hudson Mountains imply that several hundred meters of ice sheet thinning occurred during the early- to mid-Holocene (\sim 9–5 ka) in both the eastern and central ASE (Figure 2). If CDW upwelling onto the shelf occurred across a broader area of the ASE, that is, not just in Pine Island Bay, it could have caused grounding line retreat and subsequent thinning at both sites simultaneously. We therefore speculate that the similar model responses to forcing at both sites (Figure 5) and concurrent timing of thinning determined from exposure age dating (Figure 2) point to early Holocene loss of an ice shelf in the central ASE as well as in the eastern ASE.

5.5.2. Late Holocene Record of Ice Sheet Thinning

No glacial deposits or bedrock surfaces across the whole ASE have yet yielded reliable deglaciation ages younger than 5 ka (Johnson et al., 2020). The model simulations undertaken here all predict that ice surface lowering and associated grounding line retreat were essentially complete by 5 ka in both the eastern and central ASE (Figure 3 and Figures 5–8). If correct, this implies that the ice sheet surface has been stable throughout the past 5,000 years in the Hudson Mountains and at Mt Murphy, concurring with the lack of geological evidence for recent thinning, but contrasting with dramatic rates of grounding line retreat



and flow acceleration that have occurred on both Pine Island and Pope Glaciers during the past few decades (Konrad et al., 2018; Rignot et al., 2014; Shepherd et al., 2019). Alternatively, if the ice sheet has not been stable for the past 5,000 years, but instead experienced at least one episode of thinning below present and re-thickening during that time, the glacial-geological record of this would lie below the modern ice sheet surface. Such cycles have been identified in recent Antarctic-wide modeling studies (cf. Albrecht et al., 2020b; Kingslake et al., 2018; Lowry et al., 2020) and could explain why no glacial-geological record of late Holocene ice surface lowering has yet been found above the modern ice sheet surface anywhere in the ASE (Johnson et al., 2020). Glacial-geological evidence to test this hypothesis can only be obtained by subglacial bedrock recovery drilling (Spector et al., 2018).

5.5.3. Maximum LGM Ice Thickness

Reliable estimates of the maximum Antarctic ice sheet thickness at the LGM are essential for testing the validity of ice sheet models. However, thus far, no samples collected from the ASE have yielded surface exposure ages older than the LGM (none are >25 ka; most are <16 ka). Consequently, nothing is known about the maximum LGM ice sheet thickness in this sector of the Antarctic ice sheet. All model simulations except H and H' imply that the Hudson Mountains would have been just submerged or close to the ice surface at the LGM (Figure 8 and Figure S2), with the two simulations (B and D) that yield more realistic modern conditions implying submergence by ~100 m. In contrast, all simulations imply that the highest accessible exposed rock surface at Mt Murphy would have been at least ~400 m above the ice surface through the LGM. Identification of the precise location of the LGM limit on both peaks, and sampling for surface exposure dating along a transect of all ice-free surfaces above that, would help to resolve current discrepancies in model simulations of the LGM ice sheet configuration (see Section 5.4 (7)).

6. Conclusions and Further Work

A limited-area nested ice sheet model (Pollard & DeConto, 2012a) was applied to the last deglacial retreat of the West Antarctic Ice Sheet in the ASE, at 5 km resolution. The ice sheet response to ocean forcing was examined at the Hudson Mountains, Mt Murphy, and along the flowlines of the adjacent Pine Island and Pope Glaciers. By comparing model outputs with thinning histories derived from cosmogenic nuclide exposure dating, we draw the following conclusions:

- The model simulations suggest that the glacial-geological records of ice surface lowering at both sites reflect the pace and magnitude of downstream grounding line retreat.
- The model simulations predict a broadly similar ice sheet response to forcing at both sites, with an initial
 rapid phase of ice surface lowering followed by a slower phase to the modern configuration. Although the
 timing of modeled thinning prior to the Holocene contrasts with the early- to mid-Holocene ice surface
 lowering suggested by the exposure age data, the similar response at both sites is consistent with a hypothesis that both Pine Island and Pope Glaciers were responding to similar (ocean) forcing mechanisms.
- The model simulations predict several hundred meters of surface lowering at both sites. This is consistent with the exposure age data, which suggest at least 140 and 560 m of ice surface lowering during the early- to mid-Holocene in the Hudson Mountains and at Mt Murphy, respectively.
- There is a mismatch between the timing of ice surface lowering recorded by the exposure age data at both sites and the timing of thinning predicted by the model: depending on how assumptions are applied in the exposure age calculations (relating to snow cover and GIA, for example), the initial rapid phase of thinning occurs up to ~5,000 years too early in the model. With the model in its present form, this discrepancy is not resolved by changing the role of ocean melting, faster bedrock rebound, or by regional sea-level variations (see Supporting Information).

Further research would be valuable for improving both the ice sheet model and our knowledge of the pace of the last deglacial retreat in the ASE region. Although our experiments suggest that the role of climate forcing is secondary to sea-level forcing (Section 5.4 (3)), the simple prescribed atmospheric and oceanic forcing used here may not capture regional variations over the whole ASE region (such as small-scale changes in CDW influence on grounding line retreat), and should therefore be improved. Obtaining a data set of measured, rather than modeled, ocean temperatures for the Holocene in the ASE would improve this situation. High-resolution regional atmospheric and oceanic climate models would be an alternative,



but long-term simulations over the last ~20,000 years are not available to our knowledge. The model-data mismatch described here could be improved by better representations of Earth-ice interactions which could potentially change the timing of the model response to forcing. For example, global Earth-sea-level models incorporating laterally varying Earth properties (Powell et al., 2020) could be coupled to the ice sheet model to capture regional sea-surface geoid variations as well as the effects of spatially variable weak mantle viscosities below the ASE (Gomez et al., 2018). Information on spatial constraints in basal sliding properties on the continental shelf would also be valuable, and would have the potential to produce distinct model responses in different locations across the ASE. Lastly, improved glacial-geological constraints on LGM ice thicknesses in the ASE coastal region would help to resolve inter-model differences (cf. Albrecht et al., 2020a, 2020b) and further constrain basal sliding coefficients.

In terms of improving the glacial-geological data set, rock samples from a broader elevation range in the Hudson Mountains would help determine whether the peaks were submerged by ice at the LGM. A subglacial bedrock recovery drilling campaign in the Hudson Mountains forms part of the International Thwaites Glacier Collaboration (2018–2023), and will investigate late Holocene history of grounding line change in the eastern ASE. Finally, exposure dating of glacially eroded surfaces from the highest accessible ridge of Mt Murphy would provide a valuable constraint on maximum LGM thickness of the ice sheet in the ASE.

Data Availability Statement

The ice sheet model is described in Pollard and DeConto (2012a), Pollard et al. (2015), and associated supplementary materials. Selected model output files, metadata and model code are available on Penn State Data Commons archive at http://www.datacommons.psu.edu/commonswizard/MetadataDisplay.aspx?Dataset=6253 and https://doi.org/10.26208/z0m8-ez86.

Acknowledgments

The authors thank Greg Balco, James Smith, and Richard Jones for discussion, and Laura Gerrish for drafting Figure 1. The work was funded by the UK Natural Environment Research Council (grants NE/K012088/1 and NE/ K011278/1) and contributes to British Antarctic Survey's "Polar Science for Planet Earth" program. Datasets for this research are available in this data citation reference: Johnson et al. (2019b) [with Open Government Licence v.3, no access restrictions], and included in these papers and their supplementary materials: Johnson et al. (2014; 2020). The authors thank three anonymous reviewers for their thorough and constructive comments.

References

- Albrecht, T., Winkelmann, R., & Levermann, A. (2020a). Glacial-cycle simulations of the Antarctic Ice Sheet with the Parallel Ice Sheet Model (PISM) - Part 1: Boundary conditions and climatic forcing. *The Cryosphere*, 14, 599–632. https://doi.org/10.5194/tc-14-599-2020
 Albrecht, T., Winkelmann, R., & Levermann, A. (2020b). Glacial-cycle simulations of the Antarctic Ice Sheet with the Parallel Ice Sheet
- Model (PISM) Part 2: Parameter ensemble analysis. *The Cryosphere*, *14*, 633–656. https://doi.org/10.5194/tc-14-633-2020 Allevropoulos-Borrill, A. V., Nias, I. J., Payne, A. J., Golledge, N. R., & Bingham, R. J. (2020). Ocean-forced evolution of the Amundsen Sea
- catchment, West Antarctica, by 2100. *The Cryosphere*, 14, 1245–1258. https://www.the-cryosphere.net/14/1245/2020/tc-14-1245-2020. html
- Alley, R. B., & Whillans, I. M. (1984). Response of the East Antarctica Ice Sheet to sea-level rise. *Journal of Geophysical Research*, 89(C4), 6487–6493. https://doi.org/10.1029/jc089ic04p06487

An, M., Wiens, D. A., Zhao, Y., Feng, M., Nyblade, A. A., Kanao, M., et al. (2015). S-velocity model and inferred Moho topography beneath the Antarctic Plate from Rayleigh waves. *Journal of Geophysical Research Solid Earth*, *120*, 359–383. https://doi.org/10.1002/2014jb011332

Arndt, J. E., Schenke, H. W., Jakobsson, M., Nitsche, F. O., Buys, G., Goleby, B., et al. (2013). The International Bathymetric Chart of the Southern Ocean (IBCSO) Version 1.0-A new bathymetric compilation covering circum-Antarctic waters. *Geophysical Research Letters*, 40, 3111–3117. https://doi.org/10.1002/grl.50413

Arthern, R. J., & Williams, C. R. (2017). The sensitivity of West Antarctica to the submarine melting feedback. *Geophysical Research Letters*, 44, 2352–2359. https://doi.org/10.1002/2017GL072514

- Balco, G. (2011). Contributions and unrealized potential contributions of cosmogenic-nuclide exposure dating to glacier chronology, 1990-2010. Quaternary Science Reviews, 30, 3–27. https://doi.org/10.1016/j.quascirev.2010.11.003
- Balco, G., Stone, J. O., Lifton, N. A., & Dunai, T. J. (2008). A complete and easily accessible means of calculating surface exposure ages or erosion rates from ¹⁰Be and ²⁶Al measurements. *Quaternary Geochronology*, 3, 174–195. https://doi.org/10.1016/j.quageo.2007.12.001
- Balco, G., Todd, C., Goehring, B. M., Moening-Swanson, I., & Nichols, K. (2019). Glacial geology and cosmogenic-nuclide exposure ages from the Tucker Glacier - Whitehall Glacier confluence, northern Victoria Land, Antarctica. *American Journal of Science*, 319(4), 255– 286. https://doi.org/10.2475/04.2019.01
- Balco, G., Todd, C., Huybers, K., Campbell, S., Vermeulen, M., Hegland, M., et al. (2017). Cosmogenic nuclide exposure ages from the Pensacola mountains adjacent to the foundation ice stream, Antarctica. *American Journal of Science*, *316*, 542–577. https://doi.org/10.2475/06.2016.02
- Barletta, V. R., Bevis, M., Smith, B. E., Wilson, T., Brown, A., Bordoni, A., et al. (2018). Observed rapid bedrock uplift in Amundsen Sea Embayment promotes ice-sheet stability. *Science*, 360, 1335–1339. https://doi.org/10.1126/science.aa01447
- Bentley, M. J., Hein, A. S., Sugden, D. E., Whitehouse, P. L., Shanks, R., Xu, S., & Freeman, S. P. H. T. (2017). Deglacial history of the Pensacola mountains, Antarctica from glacial geomorphology and cosmogenic nuclide surface exposure dating. *Quaternary Science Reviews*, 158, 58–76. https://doi.org/10.1016/j.quascirev.2016.09.028
- Bentley, M. J., Johnson, J. S., Hodgson, D. A., Dunai, T., Freeman, S. P. H. T., Cofaigh, C. O. (2011). Rapid deglaciation of Marguerite Bay, western Antarctic Peninsula in the Early Holocene. *Quaternary Science Reviews*, 30, 3338–3349. https://doi.org/10.1016/j.quascirev.2011.09.002
- Bindschadler, R., Vornberger, P., Fleming, A., Fox, A., Mullins, J., Binnie, D., et al. (2008). The landsat image mosaic of Antarctica. *Remote Sensing of Environment*, 112, 4214–4226. https://doi.org/10.1016/j.rse.2008.07.006



- Briggs, R. D., Pollard, D., & Tarasov, L. (2014). A data-constrained large ensemble analysis of Antarctic evolution since the Eemian. Quaternary Science Reviews, 103, 91–115. https://doi.org/10.1016/j.quascirev.2014.09.003
- Briggs, R., Pollard, D., & Tarasov, L. (2013). A glacial systems model configured for large ensemble analysis of Antarctic deglaciation. The Cryosphere, 7, 1949–1970. https://doi.org/10.5194/tc-7-1949-2013
- Cuffey, K. M., Clow, G. D., Steig, E. J., Buizert, C., Fudge, T. J., Koutnik, M., et al. (2016). Deglacial temperature history of West Antarctica. Proceedings of the National Academy of Sciences of the United States of America, 113, 14249–14254. https://doi.org/10.1073/ pnas.1609132113
- DeConto, R. M., & Pollard, D. (2016). Contribution of Antarctica to past and future sea-level rise. Nature, 531, 591–597. https://doi. org/10.1038/nature17145
- Durand, G., Gagliardini, O., Favier, L., Zwinger, T., & le Meur, E. (2011). Impact of bedrock description on modeling ice sheet dynamics. Geophysical Research Letters, 38, L20501. https://doi.org/10.1029/2011GL048892
- Feldmann, J., & Levermann, A. (2015). Collapse of the West Antarctic Ice Sheet after local destabilization of the Amundsen Basin. Proceedings of the National Academy of Sciences of the United States of America, 112(46), 14191–14196. https://doi.org/10.1073/pnas.1512482112Fretwell, P., Pritchard, H. D., Vaughan, D. G., Bamber, J. L., Barrand, N. E., Bell, R., et al. (2013). Bedmap2: Improved ice bed, surface and
- thickness datasets for Antarctica. The Cryosphere, 7, 375–393. https://doi.org/10.5194/tc-7-375-2013
 Glasser, N. F., Davies, B. J., Carrivick, J. L., Rodés, A., Hambrey, M. J., Smellie, J. L., & Domack, E. (2014). Ice-stream initiation, duration and thinning on James Ross Island, northern Antarctic Peninsula. *Quaternary Science Reviews*, 86, 78–88. https://doi.org/10.1016/j. quascirev.2013.11.012
- Goehring, B. M., Balco, G., Todd, C., Moening-Swanson, I., Nichols, K. (2019). Late-glacial grounding line retreat in the northern Ross Sea, Antarctica. Geology, 474 (4), 291–294. http://dx.doi.org/10.1130/g45413.1
- Gohl, K., Uenzelmann-Neben, G., Larter, R. D., Hillenbrand, C.-D., Hochmuth, K., Kalberg, T., et al. (2013). Seismic stratigraphic record of the Amundsen Sea Embayment shelf from pre-glacial to recent times: Evidence for a dynamic West Antarctic ice sheet. *Marine Geology*, 344, 115–131. https://doi.org/10.1016/j.margeo.2013.06.011
- Golledge, N. R., Keller, E. D., Gomez, N., Naughten, K. A., Bernales, J., Trusel, L. D., & Edwards, T. L. (2019). Global environmental consequences of twenty-first-century ice-sheet melt. *Nature*, 566, 65–72. https://doi.org/10.1038/s41586-019-0889-9
- Golledge, N. R., Levy, R. H., McKay, R. M., Fogwill, C. J., White, D. A., Graham, A. G. C., et al. (2013). Glaciology and geological signature of the Last Glacial Maximum Antarctic ice sheet. *Quaternary Science Reviews*, 78, 225–247. https://doi.org/10.1016/j.quascirev.2013.08.011 Golledge, N. R., Menviel, L., Carter, L., Fogwill, C. J., England, M. H., Cortese, G., & Levy, R. H. (2014). Antarctic contribution to meltwater
- pulse 1A from reduced Southern Ocean overturning. *Nature Communications*, 5, 5107. https://doi.org/10.1038/ncomms6107 Gomez, N., Latychev, K., & Pollard, D. (2018). A coupled ice sheet-sea level model incorporating 3D Earth structure: Variations in Antarc-
- tica during the last deglacial retreat. Journal of Climate, 31, 4041–4054. https://doi.org/10.1175/jcli-d-17-0352.1
- Gomez, N., Pollard, D., & Holland, D. (2015). Sea-level feedback lowers projections of future Antarctic Ice-Sheet mass loss. Nature Communications, 6, 8798. https://doi.org/10.1038/ncomms9798
- Gomez, N., Pollard, D., & Mitrovica, J. X. (2013). A 3-D coupled ice sheet Sea level model applied to Antarctica through the last 40 ky. *Earth and Planetary Science Letters*, 384, 88–99. https://doi.org/10.1016/j.epsl.2013.09.042
- Graham, A. G. C., Larter, R. D., Gohl, K., Hillenbrand, C.-D., Smith, J. A., & Kuhn, G. (2009). Bedform signature of a West Antarctic palaeo-ice stream reveals a multi-temporal record of flow and substrate control. *Quaternary Science Reviews*, 28, 2774–2793. https://doi. org/10.1016/j.quascirev.2009.07.003
- Heeszel, D. S., Wiens, D. A., Anandakrishnan, S., Aster, R. C., Dalziel, I. W. D., Huerta, A. D., et al. (2016). Upper mantle structure of central and West Antarctica from array analysis of Rayleigh wave phase velocities. *Journal of Geophysical Research Solid Earth*, 121, 1758–1775. https://doi.org/10.1002/2015JB012616
- Hillenbrand, C.-D., Smith, J. A., Hodell, D. A., Greaves, M., Poole, C. R., Kender, S., et al. (2017). West Antarctic Ice Sheet retreat driven by Holocene warm water incursions. *Nature*, 547, 43–48. https://doi.org/10.1038/nature22995
- Jeong, A., Lee, J. I., Seong, Y. B., Balco, G., Yoo, K.-C., Yoon, H. I., et al. (2018). Late Quaternary deglacial history across the Larsen B embayment, Antarctica. Quaternary Science Reviews, 189, 134–148. https://doi.org/10.1016/j.quascirev.2018.04.011
- Johnson, J. S., Bentley, M. J., Smith, J. A., Finkel, R. C., Rood, D. H., Gohl, K., et al. (2014). Rapid Thinning of Pine Island Glacier in the Early Holocene. *Science*, 343(6174), 999–1001. https://doi.org/10.1126/science.1247385
- Johnson, J. S., Nichols, K. A., Goehring, B. M., Balco, G., & Schaefer, J. M. (2019a). Abrupt mid-Holocene ice loss in the western Weddell Sea Embayment of Antarctica. *Earth and Planetary Science Letters*, 518, 127–135. https://doi.org/10.1016/j.epsl.2019.05.002
- Johnson, J. S., Roberts, S. J., Rood, D. H., Pollard, D., Schaefer, J. M., Whitehouse, P. L., et al. (2020). Deglaciation of Pope Glacier implies widespread early Holocene ice sheet thinning in the Amundsen Sea sector of Antarctica. *Earth and Planetary Science Letters*, 548, 116501. https://doi.org/10.1016/j.epsl.2020.116501
- Johnson, J. S., Roberts, S. J., Rood, D. H., Schaefer, J. M., & Smith, J. A. (2019b). ¹⁰Be surface exposure age data from Mt Murphy, Marie Byrd Land, Antarctica (Version 1.0) UK Polar Data Centre. Natural Environment Research Council, UK Research & Innovation. https://doi. org/10.5285/63bbaf4f-fe53-45a8-b2b5-d41adcf3d35f
- Johnson, J. S., Smith, J. A., Schaefer, J. M., Young, N. E., Goehring, B. M., Hillenbrand, C.-D., et al. (2017). The last glaciation of Bear Peninsula, central Amundsen Sea Embayment of Antarctica: Constraints on timing and duration revealed by in situ cosmogenic ¹⁴C and ¹⁰Be dating. *Quaternary Science Reviews*, 178, 77–88. https://doi.org/10.1016/j.quascirev.2017.11.003
- Jones, R. S., Mackintosh, A. N., Norton, K. P., Golledge, N. R., Fogwill, C. J., Kubik, P. W., et al. (2015). Rapid Holocene thinning of an East Antarctic outlet glacier driven by marine ice sheet instability. *Nature Communications*, 6, 8910. https://doi.org/10.1038/ncomms9910
- Jones, R. S., Norton, K. P., Mackintosh, A. N., Anderson, J. T. H., Kubik, P., Vockenhuber, C., et al. (2017). Cosmogenic nuclides constrain surface fluctuations of an East Antarctic outlet glacier since the Pliocene. *Earth and Planetary Science Letters*, 480, 75–86. https://doi. org/10.1016/j.epsl.2017.09.014
- Jones, R. S., Small, D., Cahill, N., Bentley, M. J., & Whitehouse, P. L. (2019a). iceTEA: Tools for plotting and analyzing cosmogenic-nuclide surface-exposure data from former ice margins. *Quaternary Geochronology*, 51, 72–86. https://doi.org/10.1016/j.quageo.2019.01.001
- Jones, R. S., Whitehouse, P. L., Bentley, M. J., Small, D., & Dalton, A. S. (2019b). Impact of glacial isostatic adjustment on cosmogenic surface-exposure dating. *Quaternary Science Reviews*, 212, 206–212. https://doi.org/10.1016/j.quascirev.2019.03.012
- Kingslake, J., Scherer, R. P., Albrecht, T., Coenen, J., Powell, R. D., Reese, R., et al. (2018). Extensive retreat and re-advance of the West Antarctic Ice Sheet during the Holocene. *Nature*, 558, 430–434. https://doi.org/10.1038/s41586-018-0208-x
- Kirshner, A. E., Anderson, J. B., Jakobsson, M., O'Regan, M., Majewski, W., & Nitsche, F. O. (2012). Post-LGM deglaciation in Pine Island Bay, West Antarctica. *Quaternary Science Reviews*, 38, 11–26. https://doi.org/10.1016/j.quascirev.2012.01.017



- Konrad, H., Shepherd, A., Gilbert, L., Hogg, A. E., McMillan, M., Muir, A., & Slater, T. (2018). Net retreat of Antarctic glacier grounding lines. Nature Geoscience, 11, 258–262. https://doi.org/10.1038/s41561-018-0082-z
- Larter, R. D., Anderson, J. B., Graham, A. G. C., Gohl, K., Hillenbrand, C.-D., Jakobsson, M., et al. (2014). Reconstruction of changes in the Amundsen Sea and Bellingshausen Sea sector of the West Antarctic Ice Sheet since the Last Glacial Maximum. *Quaternary Science Reviews*, 100, 55–86. https://doi.org/10.1016/j.quascirev.2013.10.016
- Le Brocq, A. M., Payne, A. J., & Vieli, A. (2010). An improved Antarctic dataset for high resolution numerical ice sheet models (ALBMAP v1). *Earth System Science Data*, *2*, 247–260. https://doi.org/10.5194/essd-2-247-2010
- Lifton, N., Sato, T., & Dunai, T. J. (2014). Scaling in situ cosmogenic nuclide production rates using analytical approximations to atmospheric cosmic-ray fluxes. *Earth and Planetary Science Letters*, 386, 149–160. https://doi.org/10.1016/j.epsl.2013.10.052
- Lisiecki, L. E., & Raymo, M. E. (2005). A Pliocene-Pleistocene stack of 57 globally distributed benthic δ18O records. *Paleoceanography*, 20, PA1003. https://doi.org/10.1029/2004PA001071
- Liu, Z., Otto-Bliesner, B. L., He, F., Brady, E. C., Tomas, R., Clark, P. U., et al. (2009). Transient simulation of last deglaciation with a new mechanism for Bolling-Allerod warming. Science, 325, 310–314. https://doi.org/10.1126/science.1171041
- Lowry, D. P., Golledge, N. R., Bertler, N. A. N., Jones, R. S., McKay, R., & Stutz, J. (2020). Geologic controls on ice sheet sensitivity to deglacial climate forcing in the Ross Embayment, Antarctica. *Quaternary Science Advances*, 1, 100002. https://doi.org/10.1016/j.qsa.2020.100002
- Lowry, D. P., Golledge, N. R., Bertler, N. A. N., Selwyn Jones, R. S., & McKay, R. (2019b). Deglacial grounding-line retreat in the Ross Embayment, Antarctica, controlled by ocean and atmosphere forcing. *Science Advances*, 5, eaav8754. https://doi.org/10.1126/sciadv. aav8754
- Lowry, D. P., Golledge, N. R., Menviel, L., & Bertler, N. A. N. (2019a). Deglacial evolution of regional Antarctic climate and Southern Ocean conditions in transient climate simulations. *Climate of the Past*, 15, 189–215. https://doi.org/10.5194/cp-15-189-2019
- Mackintosh, A., Golledge, N., Domack, E., Dunbar, R., Leventer, A., White, D., et al. (2011). Retreat of the East Antarctic ice sheet during the last glacial termination. *Nature Geoscience*, 4, 195–202. https://doi.org/10.1038/ngeo1061
- Maris, M. N. A., van Wessem, J. M., van de Berg, W. J., de Boer, B., & Oerlemans, J. (2015). A model study of the effect of climate and sea-level change on the evolution of the Antarctic Ice Sheet from the Last Glacial Maximum to 2100. *Climate Dynamics*, *45*, 837–851. https://doi.org/10.1007/s00382-014-2317-z
- Milillo, P., Rignot, E., Rizzoli, P., Scheuchl, B., Mouginot, J., Bueso-Bello, J., & Prats-Iraola, P. (2019). Heterogeneous retreat and ice melt of Thwaites Glacier, West Antarctica. Science Advances, 5(1), eaau3433. https://doi.org/10.1126/sciady.aau3433
- Mouginot, J., Scheuchl, B., & Rignot, E. (2012). Mapping of ice motion in Antarctica using synthetic-aperture radar data. *Remote Sensing*, 4, 2753–2767. https://doi.org/10.3390/rs4092753
- Nichols, K. A., Goehring, B. M., Balco, G., Johnson, J. S., Hein, A. S., & Todd, C. (2019). New Last Glacial Maximum ice thickness constraints for the Weddell Sea Embayment, Antarctica. *The Cryosphere*, 13, 2935–2951. https://doi.org/10.5194/tc-2019-6410.5194/tc-13-2935-2019
- O'Donnell, J. P., Selway, K., Nyblade, A. A., Brazier, R. A., Wiens, D. A., Anandakrishnan, S., et al. (2017). The uppermost mantle seismic velocity and viscosity structure of central West Antarctica. *Earth and Planetary Science Letters*, 472, 38–49. https://doi.org/10.1016/j. epsl.2017.05.016
- Parizek, B. R., Christianson, K., Anandakrishnan, S., Alley, R. B., Walker, R. T., Edwards, R. A., et al. (2013). Dynamic (in)stability of Thwaites Glacier, West Antarctica. Journal of Geophysical Research Surfaces, 118, 638–655. https://doi.org/10.1002/jgrf.20044
- Peltier, W. R. (2004). Global glacial isostasy and the surface of the Ice-Age Earth: The ICE-5G (VM2) Model and GRACE. Annual Review of Earth and Planetary Sciences, 32, 111–149. https://doi.org/10.1146/annurev.earth.32.082503.144359
- Peltier, W. R., Argus, D. F., & Drummond, R. (2015). Space geodesy constrains ice age terminal deglaciation: The global ICE-6G_C (VM5a) model. Journal of Geophysical Research Solid Earth, 120, 450–487. https://doi.org/10.1002/2014jb011176
- Pollard, D., Chang, W., Haran, M., Applegate, P., & DeConto, R. (2016). Large ensemble modeling of the last deglacial retreat of the West Antarctic Ice Sheet: Comparison of simple and advanced statistical techniques. *Geoscientific Model Development*, 9, 1697–1723. https:// doi.org/10.5194/gmd-9-1697-2016
- Pollard, D., & DeConto, R. M. (2009). Modeling West Antarctic ice sheet growth and collapse through the past five million years. *Nature*, 458, 329–332. https://doi.org/10.1038/nature07809
- Pollard, D., & DeConto, R. M. (2012a). Description of a hybrid ice sheet-shelf model, and application to Antarctica. Geoscientific Model Development, 5, 1273–1295. https://doi.org/10.5194/gmd-5-1273-2012
- Pollard, D., & DeConto, R. M. (2012b). A simple inverse method for the distribution of basal sliding coefficients under ice sheets, applied to Antarctica. *The Cryosphere*, 6, 953–971. https://doi.org/10.5194/tc-6-953-2012
- Pollard, D., DeConto, R. M., & Alley, R. B. (2015). Potential Antarctic Ice Sheet retreat driven by hydrofracturing and ice cliff failure. Earth and Planetary Science Letters, 412, 112–121. https://doi.org/10.1016/j.epsl.2014.12.035
- Pollard, D., Gomez, N., & DeConto, R. M. (2017). Variations of the Antarctic Ice Sheet in a coupled ice sheet-Earth-sea level model: Sensitivity to viscoelastic Earth properties. Journal of Geophysical Research Surfaces, 122, 2124–2138. https://doi.org/10.1002/2017JF004371
- Powell, E., Gomez, N., Hay, C., Latychev, K., & Mitrovica, J. X. (2020). Viscous effects in the solid Earth response to modern Antarctic ice mass flux: Implications for geodetic studies of WAIS stability in a warming world. *Journal of Climate*, 33, 443–459. https://doi. org/10.1175/jcli-d-19-0479.1
- RAISED consortium. (2014). A community-based geological reconstruction of Antarctic Ice Sheet deglaciation since the Last Glacial Maximum. *Quaternary Science Reviews*, 100, 1–9. https://doi.org/10.1016/j.quascirev.2014.06.025
- Rignot, E. J., Mouginot, J. & Scheuchl, B. (2017). MEaSUREs InSAR-based Antarctica Ice velocity map, Version 2. Boulder, Colorado USA. NASA national snow and ice data center distributed active archive center. https://doi.org/10.5067/D7GK8F5J8M8R
- Rignot, E., Mouginot, J., Morlighem, M., Seroussi, H., & Scheuchl, B. (2014). Widespread, rapid grounding line retreat of Pine Island, Thwaites, Smith, and Kohler glaciers, West Antarctica, from 1992 to 2011. *Geophysical Research Letters*, 41, 3502–3509. https://doi. org/10.1002/2014gl060140
- Rignot, E., Mouginot, J., & Scheuchl, B. (2011a). Ice flow of the Antarctic Ice Sheet. Science, 333, 1428–1430. https://doi.org/10.1126/ science.1208336
- Rignot, E., Mouginot, J., & Scheuchl, B. (2011b). Antarctic grounding line mapping from differential satellite radar interferometry. Geophysical Research Letters, 38, L10504. https://doi.org/10.1029/2011gl047109
- Schoof, C. (2007). Ice sheet grounding line dynamics: Steady states, stability, and hysteresis. *Journal of Geophysical Research*, *112*, F03S28. https://doi.org/10.1029/2006JF000664
- Seroussi, H., Nakayama, Y., Larour, E., Menemenlis, D., Morlighem, M., Rignot, E., & Khazendar, A. (2017). Continued retreat of Thwaites Glacier, West Antarctica, controlled by bed topography and ocean circulation. *Geophysical Research Letters*, 44, 6191–6199. https://doi. org/10.1002/2017GL072910



- Shepherd, A., Gilbert, L., Muir, A. S., Konrad, H., McMillan, M., Slater, T., et al. (2019). Trends in Antarctic Ice Sheet elevation and mass. Geophysical Research Letters, 46(14), 8174–8183. https://doi.org/10.1029/2019gl082182
- Small, D., Bentley, M. J., Jones, R. S., Pittard, M. L., & Whitehouse, P. L. (2019). Antarctic ice sheet palaeo-thinning rates from vertical transects of cosmogenic exposure ages. *Quaternary Science Reviews*, 206, 65–80. https://doi.org/10.1016/j.quascirev.2018.12.024
- Smellie, J. L., Rocchi, S., Johnson, J. S., Di Vincenzo, G., & Schaefer, J. M. (2017). A tuff cone erupted under frozen-bed ice (northern Victoria Land, Antarctica): Linking glaciovolcanic and cosmogenic nuclide data for ice sheet reconstructions. *Bulletin of Volcanology*, 80, 12. https://doi.org/10.1007/s00445-017-1185-x
- Smith, J. A., Hillenbrand, C.-D., Kuhn, G., Klages, J. P., Graham, A. G. C., Larter, R. D., et al. (2014). New constraints on the timing of West Antarctic Ice Sheet retreat in the eastern Amundsen Sea since the last Glacial Maximum. *Global and Planetary Change*, 122, 224–237. https://doi.org/10.1016/j.gloplacha.2014.07.015
- Smith, J. A., Hillenbrand, C.-D., Kuhn, G., Larter, R. D., Graham, A. G. C., Ehrmann, W., et al. (2011). Deglacial history of the West Antarctic Ice Sheet in the western Amundsen Sea Embayment. *Quaternary Science Reviews*, 30, 488–505. https://doi.org/10.1016/j. quascirev.2010.11.020
- Spector, P., Stone, J., Cowdery, S. G., Hall, B., Conway, H., Bromley, G. (2017). Rapid early-Holocene deglaciation in the Ross Sea, Antarctica. *Geophysical Research Letters*, 44, 7817–7825. https://doi.org/10.1002/2017GL074216
- Spector, P., Stone, J., & Goehring, B. (2019). Thickness of the divide and flank of the West Antarctic Ice Sheet through the last deglaciation. *The Cryosphere*, 13, 3061–3075. https://doi.org/10.5194/tc-13-3061-2019
- Spector, P., Stone, J., Pollard, D., Hillebrand, T., Lewis, C., & Gombiner, J. (2018). West Antarctic sites for subglacial drilling to test for past ice-sheet collapse. *The Cryosphere*, 12, 2741–2757. https://doi.org/10.5194/tc-12-2741-2018
- Stone, J. O., Balco, G. A., Sugden, D., Caffee, M., Sass, L. S., III, Cowdery, S. G., & Siddoway, C. (2003). Holocene deglaciation of Marie Byrd Land, West Antarctica. Science, 299, 99–102. https://doi.org/10.1126/science.1077998
- Strub, E., Wiesel, H., Delisle, G., Binnie, S. A., Liermann, A, Dunai, D. J., Herpers, U., et al. (2015). Glaciation history of Queen Maud Land (Antarctica) – New exposure data from nunataks. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 361, 599–603. https://doi.org/10.1016/j.nimb.2015.04.052
- WAIS Divide Project Members. (2013). Onset of deglacial warming in West Antarctica driven by local orbital forcing. *Nature*, 500, 440–444. https://doi.org/10.1038/nature12376
- Whitehouse, P. L., Bentley, M. J., & Le Brocq, A. M. (2012). A deglacial model for Antarctica: geological constraints and glaciological modelling as a basis for a new model of Antarctic glacial isostatic adjustment. *Quaternary Science Reviews*, 32, 1–24. https://doi.org/10.1016/j. quascirev.2011.11.016
- Yokoyama, Y., Anderson, J. B., Yamane, M., Simkins, L. M., Miyairi, Y., Yamazaki, T., et al. (2016). Widespread collapse of the Ross Ice Shelf during the late Holocene. Proceedings of the National Academy of Sciences of the United States of America, 113, 2354–2359. https://doi. org/10.1073/pnas.1516908113

References From the Supporting Information

- Cornford, S. L., Martin, D. F., Lee, V., Payne, A. J., & Ng, E. G. (2016). Adaptive mesh refinement versus subgrid friction interpolation in simulations of Antarctic ice dynamics. *Annals of Glaciology*, 57, 1–9. https://doi.org/10.1017/aog.2016.13
- Cornford, S. L., Seroussi, H, Asay-Davis, X. S., Gudmundsson, G. H., Arthern, S. L., Cornford, S. L., et al. (2020). Results of the third Marine Ice Sheet Model Intercomparison Project (MISMIP+). *The Cryosphere Discussions*. https://doi.org/10.5194/tc-2019-326
- DeConto, R. M., Pollard, D., & Kowalewski, D. (2012). Modeling Antarctic ice sheet and climate variations during Marine Isotope Stage 31. Global and Planetary Change, 88–89, 45–52. https://doi.org/10.1016/j.gloplacha.2012.03.003
- Drouet, A. S., Docquier, D., Durand, G., Hindmarsh, R., Pattyn, F., Gagliardini, O., & Zwinger, T. (2013). Grounding line transient response in marine ice sheet models. *The Cryosphere*, 7, 395–406. https://doi.org/10.5194/tc-7-395-2013
- Gladstone, R. M., Lee, V., Vieli, A., & Payne, A. J. (2010). Grounding line migration in an adaptive mesh ice sheet model. Journal of Geophysical Research, 115, F04014. https://doi.org/10.1029/2009JF001615
- Gladstone, R. M., Payne, A. J., & Cornford, S. L. (2012). Resolution requirements for grounding-line modeling: Sensitivity to basal drag and ice-shelf buttressing. *Annals of Glaciology*, 53, 97–105. https://doi.org/10.3189/2012aog60a148
- Goldberg, D., Holland, D. M., & Schoof, C. (2009). Grounding line movement and ice shelf buttressing in marine ice sheets. Journal of Geophysical Research, 114, F04026. https://doi.org/10.1029/2008JF001227
- Larour, E., Seroussi, H., Adhikari, S., Ivins, E., Caron, L., Morlighem, M., & Schlegel, N. (2019). Slowdown in Antarctic mass loss from solid Earth and sea-level feedbacks. *Science*, *364*, eaav7908. https://doi.org/10.1126/science.aav7908
- Lee, B. S., Haran, M., Fuller, R., Pollard, D., & Keller, K. (2020). A fast particle-based approach for calibrating a 3-D model of the Antarctic Ice Sheet. Annals of Applied Statistics, 14, 605–634. https://doi.org/10.1214/19-aoas1305
- Pattyn, F., & Durand, G. (2013). Why marine ice sheet model predictions may diverge in estimating future sea level rise. *Geophysical Research Letters*, 40, 4316–4320. https://doi.org/10.1002/grl.50824
- Pattyn, F., Perichon, L., Durand, G., Favier, L., Gagliardini, O., Hindmarsh, R. C. A., et al. (2013). Grounding-line migration in planview marine ice-sheet models: Results of the ice2sea MISMIP3d intercomparison. *Journal of Glaciology*, 59, 410–422. https://doi. org/10.3189/2013jog12j129
- Pattyn, F., Schoof, C., Perichon, L., Hindmarsh, R. C. A., Bueler, E., de Fleurian, B., et al. (2012). Results of the Marine Ice Sheet Model Intercomparison Project, MISMIP. *The Cryosphere*, 6, 573–588. https://doi.org/10.5194/tc-6-573-2012
- Pollard, D., & DeConto, R. M. (2020). Improvements in one-dimensional grounding-line parameterizations in an ice-sheet model with lateral variations (PSUICE3D v2.1). Geoscientific Model Development, 13, 6481–6500. https://doi.org/10.5194/gmd-13-6481-2020
- Whitehouse, P. L. (2018). Glacial isostatic adjustment modeling: Historical perspectives, recent advances, and future directions. *Earth Surface Dynamics*, 6, 401–429. https://doi.org/10.5194/esurf-6-401-2018